Author’s response to reviews

Title: A Novel Mutation in SEPN1 Causing Rigid Spine Muscular Dystrophy 1: A Case Report

Authors:

Fateme Ziyaee (ziyaeefateme@yahoo.com)
Eslam Shorafa (omidshorafa@gmail.com)
Hassan Dastsooz (genetichf@gmail.com)
Parham Habibzadeh (parham.habibzadeh@yahoo.com)
Hamid Nemati (nematih@sums.ac.ir)
Amir Saeed (dr.saeedamir@yahoo.com)
Mohammad Silawi (silawimohammad@yahoo.com)
Mohammad Ali Frazi Fard (hamedfiroozi66@gmail.com)
Mohammad Ali Faghihi (MFaghihi@med.miami.edu)
Seyed Alireza Dastgheib (dastgheib@sums.ac.ir)

Version: 3 Date: 19 Dec 2018

Author’s response to reviews:

Dear Editor,

Thank you very much for considering our manuscript for publication in BMC medical genetics. We revised the manuscript according to your comments. In the resubmitted revision point to point responses to every point raised are provided. I hope you will find the revised version acceptable for publication in your prestigious journal.

Authors’ response to the comments:

1. “Please can you complete and send in a new Authorship Change Form, we cannot accept Authorship Change Forms where the authors have not signed the form, it appears that a lot
of the signatures were scanned onto the form. If all of the authors cannot sign one form, multiple forms can be signed and sent in.”

The new Authorship Change Form is uploaded.

2. “Please have the text edited by a professional language editing service or a native English speaking colleague”

The manuscript was edited.

3. “We would like to ask for you to provide more justification for the contributions of all authors”

Authors’ contributions were updated.

Thank you again for the careful review and suggestive comments.

Sincerely, Seyed Alireza Dastgheib, MD, MPH, PhD on behalf of all authors